Author's response to reviews

Title: Do adverse childhood experiences increase the risk of postdeployment posttraumatic stress disorder in US Marines?

Authors:

Cynthia A LeardMann (cynthia.leardmann@med.navy.mil)
Besa Smith (besa.smith@med.navy.mil)
Margaret AK Ryan (margaret.ryan@med.navy.mil)

Version: 5 Date: 22 July 2010

Author's response to reviews:

Dear Editor Norton,

Enclosed please find our revised manuscript titled, “Do adverse childhood experiences increase the risk of postdeployment posttraumatic stress disorder in US Marines?”

We have made only a few minor grammatical edits. They include the following:
1) removed comma after factors in the background section of the abstract
2) Ref 19 capitalized "Assessment"
3) Ref 21 changed "Jama" to "JAMA"
4) Ref 26 removed one of the "Alcohol" journal titles
5) Ref 35 Changed "assault posttraumatic stress disorder after combat deployment" to lower case
6) Table 4 expanded second row so reader could see the word "categories"

Thank you publishing our article in the BMC Public Health. As corresponding author, please contact me if you have any additional concerns or questions.

Very Respectfully,

Cynthia LeardMann, MPH
Biostatistician, DoD Center for Deployment Health Research
Naval Health Research Center, San Diego, CA